Author's response to reviews

Title: Sero-prevalence and risk factors for hepatitis B virus infection among health care workers in a tertiary hospital in Uganda

Authors:

Abdhalah K Ziraba (akziraba@yahoo.com)
Josephine Bwogi (josephinebwn@yahoo.co.uk)
Alice Namale (asnamal@hotmail.com)
Caroline W Wainaina (carowangoi@yahoo.com)
Harriet Mayanja-Kizza (hmk@chs.mak.ac.ug)

Version: 10 Date: 28 June 2010

Author's response to reviews:

Dear Editor,

We have proof read and carried out minor edits in the manuscript particularly replacing "hepatitis B infection" with "hepatitis B virus infection". We are certified that the manuscript is ready for publication.

Abdhalah Ziraba

(Corresponding author)